CENWP-OP-B                                                                                                                                     23 March 2003

MEMORANDUM FOR CENWP-OP, ATTN: Gary Johnson

SUBJECT: Bonneville Lock and Dam, Fishway and Fish Activities for week 12 of 2003, which covers the period 10 through 22 March 2003.

1.  OPERATION SUMMARY.

1.1 Daily river flow averaged 157.8 to 198.7 kcfs.  Daily project tailwater elevation averaged 15.3’ to 17.2’ msl.  Daily project forebay elevation averaged 74.8’ to 75.4’ msl.  River temperature averaged 41 to 43° F.  Daily turbidity measured 4.0’ to 5.0’ on the secchi disk.

1.2 Spill Bays 1 and 18, which supply attraction flow for Cascade Island and Bradford Island fishways respectively, are open .5’ during daylight hours in accordance to the FPP.  Spill averaged 1.2 to 1.9 kcfs.

1.3 Unit Operation

1.3.1 Unit 1 remains out of service for turbine rehab.

1.3.2 Unit 16 was out of service from 2342 on 17 March until 1006 on 18 March due to a STS failure.

1.3.3 Unit 17 went out of service at 1430 on 22 March due to a speed adjuster failure.  

1.3.4 Unit 18 was out of service from 1048 until 1200 on 18 March to facilitate repairs to USGS antennas.

1.4 Second Powerhouse fish units provide attraction flow for the Washington Shore fish ladder.  During fish passage season, the units are placed on standby from approximately 0000-0300 nightly if drawdown exceeds 1.0’.  Fish Unit 1 remains out of service for overhaul and stator rewind.

1.5 Gatewell drawdowns are measured on Wednesdays.

2.  ACTIVITIES. 
2.1 AWS Closures.  The following AWS valves were closed for cleaning this reporting week.

FV3-7 closed from 1240 until 1346 on 18 March.

2.2 STS/VBS Inspections.  No STS/VBSs inspections this reporting week. 

2.3 Dewatering and Fish Salvage.  On 21 March, the PDS was dewatered for contractor work.

3.  RESEARCH.

3.1 University of Idaho/WDFW jaw tagging.  Steelhead and spring chinook will be tagged for studies involving the tooth tangle net fishery on the Columbia.

3.2 NMFS Lamprey.  Tracking continues once a month.

3.3 ODFW Pikeminnow Tagging.  Electrofishing in Bonneville’s forebay.

3.4 PSMFC.  Smolt monitoring continues.  74 steelhead, three salmon and one smolt slid off the end of the separator bars.  

3.5 USGS Juvenile Radiotracking.  USGS began installing equipment around the project.

3.6 WES/Battelle/Mevatech Juvenile hydroacoustic monitoring.  Installing equipment.

3.7 WDFW Fish Counting.  16-hr video counts at Washington Shore and Bradford Island.

3.8 WDFW steelhead.  Trapping and tagging of winter steelhead continues.

4.  FISHWAYS 
4.1 Project biologists inspected 16-22 March this week.

4.2 Fish Passage Plan Observations.

	DATE
	LOCATion
	FPP Deviation
	CAUSE
	RESPONSE
	TR/WAR

	All week
	PH1 PCC
	Velocity <1.5fps
	Meter in slowest part of channel
	No response at this time.
	

	All week
	PH2 PCC
	No velocity measure.
	Meter is missing
	See misc. 5.1
	

	3/16-17
	PH2 South monolith
	Differential <1.0’
	One fish unit in operation.
	See misc. 5.5
	


4.3 Fish Ladder Temperatures.  Temperatures are not yet being recorded because we are awaiting new probe attachments.  

4.4 Sensor Calibration.  Calibration checked on 20 March.

4.5  Zebra Mussel Monitoring.  No signs of colonization were observed this reporting week. 

5. MISCELLANEOUS

5.1 Divers discovered the PH2 velocity meter was missing from the collection channel.  A new meter will be installed during the mid season dives this summer.

5.2 Cascade Island controls are in manual mode awaiting tailwater sensor calibration.  Electricians are awaiting software.  B-Branch is operating in auto. 

5.3 On 24 January, the Project tailwater hard constraint decreased to 11.0’ msl for all hours.  The soft constraint went to 11.2’ msl between 0700-1900 hours.  This is for the protection of the chum redds around Ives Island.  On 17 February, BPA agreed to a hard tailwater constraint of 11.3’ msl and a soft constraint of 11.5’ msl when inflows to Bonneville are >120Kcfs.

5.4 At 2240 on 02 March, the PH1 orifice and sluice gates were closed.  Bulkheads will be installed later this spring.

5.5 The Project is currently operating under new guidelines for one fish unit operation.  Even with the mods to the floating orifice gates, we still have some trouble maintaining differential at the south end.  We will manipulate entrance gates to try to achieve proper differential.

5.6 The manual orifices at Fish Unit 2 were closed at 1500 on 16 March.  They had been opened to keep channel elevations in criteria during the low forebay restriction last week.

5.7 The main dam was on seal because of dive operations off Bradford Island on 17-21 March.  Sediment samples were taken as part of the Bradford Island Superfund cleanup.

5.8 Debarking activities in PH2 gatewells were conducted on 17 and 20 March.

5.9 The navlock returned to service on 20 March. 

5.10 Unit 11 operated outside 1% efficiency from 1000 until 1612 on 20 March to facilitate trash rack raking.

5.11 The PDS was dewatered from 0800 until 1745 on 21 March to facilitate realignment of the screen cleaner rail.

5.12 Smolt monitoring researchers have reported a higher than average descaling rate.  This is believed to be due to debris in the forebay and gatewells.  We are currently looking into getting a tug to move the debris from PH2 forebay out through the navlock.

5.13 Sea lions and harbor seals have been observed in the tailrace of both powerhouses and the spillway.  No action is being taken by the project to harass, harm, or otherwise annoy these animals.  FFU is monitoring and recording their activity.

5.14 DSM1 is running south as we are waiting for all clearances to be released.  We expect to return the DSM to normal flow on Monday, 24 March.  

 JAMES R. MAHAR

                                 

                        

                                   Operations Manager
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